Author's response to reviews

Title: Tanzanian lessons in using non-physician clinicians to scale up comprehensive emergency obstetric care in remote and rural areas

Authors:

Angelo S Nyamtema (nyamtema_angelo@yahoo.co.uk)
Senga K Pemba (pemba@healthtrainingifakara.org)
Godfrey Mbaruku (gmbaruku@ihi.or.tz)
Fulgence D Rutasha (dadi@unfpa.org)
Jos van Roosmalen (J.J.M.van_Roosmalen@lumc.nl)

Version: 6 Date: 7 November 2011

Author's response to reviews:

Minor Corrections- typographical errors
1. Cover page: A full stop has been added after a letter K in the name of Senga K. Pemba
2. Page 6, paragraph 1: words written in a comment have been deleted
3. Page 11, para 3: A full stop has been added at the end of the last sentence.
4. Reference No. 4: Last name is Murray C.
5. Reference No 21: Space had been added between a year and vol of the article.